\documentclass[12pt]{amsart}

\usepackage{amssymb,amsthm,amsmath}
\usepackage[numbers,sort&compress]{natbib}
\usepackage[paper=a4paper, left=2.35cm, right=2.35cm, top=2.5cm, bottom=1.8cm, headheight=5.5pt, footskip=0.8cm, footnotesep=0.8cm, centering, includefoot]{geometry}
%\usepackage{showkeys}

\title[A fourth order problem with variable exponent]{Combined effects of concave-convex nonlinearities in a fourth order problem with variable exponent}

\author{Sami Baraket}
\address[Sami Baraket]{Department of Mathematics, College of Science,
King Saud University, P.O. Box 2455,
Riyadh 11451, Saudi Arabia}
\email{sbaraket@ksu.edu.sa}

\author{Vicen\c tiu D. R\u adulescu}
\address[Vicen\c tiu D. R\u adulescu]{Institute of Mathematics ``Simion Stoilow" of the Romanian Academy, P.O. Box 1-764, 014700 Bucharest, Romania \& Department of Mathematics, University of Craiova, 200585 Craiova, Romania}
\email{vicentiu.radulescu@math.cnrs.fr}


%\subjclass{Primary ; Secondary , }

\keywords{nonhomogeneous elliptic problem; variable exponent; Dirichlet boundary condition; Ekeland variational principle.}
\thanks{{\em 2010 Mathematics Subject Classification.} Primary: 35J30. Secondary: 35B38; 35J35; 35J40; 58E30.}

\newcommand{\dint}[4]{\displaystyle\int_{#1}^{#2} \! #3 \, \operatorname{d} \! #4}
\newcommand{\dd}{\,\mathrm{d}}
\newcommand{\norm}[1]{\left\| #1 \right\|}
\newcommand{\abs}[1]{\left\lvert #1 \right\rvert}
\newcommand{\RR}{\mathbb{R}}
\newcommand{\intom}{\int_\Omega}
\newcommand{\di}{\displaystyle}
\newcommand{\ep}{\varepsilon}
\newcommand{\ri}{\rightarrow}
\newcommand{\dpx}{\Delta_{p(x)}}
\newcommand{\sob}{W^{1,p(x)}(\Omega)}
\newcommand{\bb}{\begin{equation}}
\newcommand{\bbb}{\end{equation}}
\newcommand{\ee}{{\mathcal E}_\lambda}
\newcommand{\jj}{{\mathcal J}_\lambda}
\DeclareMathOperator{\essinf}{essinf}
\DeclareMathOperator{\meas}{meas}
\DeclareMathOperator{\divv}{div}
\DeclareMathOperator{\spann}{span}

\theoremstyle{plain}
\newtheorem{theorem}{Theorem}[section]
\newtheorem{corollary}[theorem]{Corollary}
\newtheorem{lemma}[theorem]{Lemma}
\newtheorem{proposition}[theorem]{Proposition}

\theoremstyle{definition}
\newtheorem{definition}[theorem]{Definition}
\newtheorem{conjecture}[theorem]{Conjecture}
\newtheorem{example}[theorem]{Example}
\newtheorem{remark}[theorem]{Remark}
\newtheorem*{note}{Note}
\newtheorem{case}{Case}[section]
\begin{document}
\begin{thebibliography}{99}
{\small
\bibitem{abc} A. Ambrosetti, H. Brezis, G. Cerami, Combined effects of concave and convex nonlinearities in some elliptic problems,
{\it J. Funct. Anal.} {\bf 122} (1994), 519-543.

\bibitem{ambrab} A.  Ambrosetti, P.H.  Rabinowitz,  Dual  variational  methods  in  critical point  theory, {\it J. Funct. Anal.}
{\bf 14} (1973), 349-381.

\bibitem{autuori} G. Autuori, F. Colasuonno, P. Pucci, 
On the existence of stationary solutions for higher-order $p$-Kirchhoff
problems, {\it Commun. Contemp. Math.} {\bf 16} (2014), no. 5, 1450002, 43 pp.

\bibitem{brezis} H. Brezis, {\it Analyse Fonctionnelle.  Th\'eorie et Applications} (French) [Functional Analysis, Theory and Applications], Collection Math\'ematiques Appliqu\'ees pour la Ma\^{\i}trise [Collection of Applied Mathematics for the Master's Degree] Masson, Paris, 1983.

\bibitem{brenir} H. Brezis, L. Nirenberg, Remarks on finding critical points, {\it Comm. Pure Appl. Math.} {\bf 44} (1991), 939-963.
    
\bibitem{chen} Z. Chen,  Infinitely many solutions to $p(x)$-biharmonic
problem with Navier boundary conditions, {\it Ann. Differential Equations} {\bf
30} (2014),  272-281.

\bibitem{cola} F. Colasuonno, P. Pucci, Multiplicity of
solutions for $p(x)$-polyharmonic elliptic Kirchhoff equations, {\it
Nonlinear Anal.} {\bf 74} (2011),  5962-5974.

\bibitem{cruz}   D. Cruz-Uribe, A. Fiorenza, {\it
Variable Lebesgue Spaces:  Foundations and Harmonic Analysis},
Applied and Numerical Harmonic Analysis, Birkh\"auser, Basel, 2013.

\bibitem{die} L. Diening, P. H\"asto, P. Harjulehto, M. Ruzicka, {\it Lebesgue and Sobolev Spaces with Variable Exponents}, Springer Lecture Notes, vol. 2017, Springer-Verlag, Berlin, 2011.

\bibitem{eke} I. Ekeland, On the variational principle, {\it J. Math. Anal. Appl.} {\bf 47} (1974), 324-353.

\bibitem{ferrara} M. Ferrara, G. Molica Bisci, D. Repov\v{s}, Existence results for nonlinear elliptic problems on fractal domains, {\it Adv. Nonlinear Anal.}, in press (DOI: 10.1515/anona-2015-0105).

\bibitem{garcia} J. Garcia Azorero, I. Peral Alonso, Multiplicity of solutions for elliptic problems with critical exponent or with a nonsymmetric term, {\it Trans. Amer. Math. Soc.} {\bf 323} (1991), 877-895.

\bibitem{gazrad} F. Gazzola, V. R\u adulescu, A nonsmooth critical point theory approach to some nonlinear elliptic equations in unbounded domains, {\it Diff. Integral Equations} {\bf 13} (2000), 47-60.

\bibitem{kong} L. Kong, Eigenvalues for a fourth order elliptic problem, {\it Proc. Amer. Math. Soc.} {\bf 143} (2015), 249-258.
    
\bibitem{krat} A. Kratohvil, I. Necas, The discreteness of the
spectrum of a nonlinear Sturm-Liouville equation of fourth order, {\it
Comment. Math. Univ. Carolinae} {\bf 12} (1971), 639-653.

\bibitem{leray} J. Leray, J.-L. Lions, Quelques r\'esultats de Visik sur les prob\`emes elliptiques non lin\'eaires par les
m\'ethodes de Minty-Browder, {\it Bull. Soc. Math. France} {\bf 93} (1965), 97-107.

\bibitem{luby} V. Lubyshev, Multiple solutions of an even-order nonlinear
problem with convex-concave nonlinearity, {\it Nonlinear Anal.} {\bf 74} (2011), 1345-1354.

\bibitem{mrpams} M. Mih\u ailescu, V.D. R\u adulescu, On a nonhomogeneous quasilinear eigenvalue problem in Sobolev spaces with variable exponent, {\it Proc. Amer. Math. Soc.} {\bf 135} (2007), 2929-2937.

\bibitem{molrep} G. Molica Bisci, D. Repov\v{s}, Multiple solutions for elliptic equations involving a general operator in divergence form, {\it Ann. Acad. Sci. Fenn. Math.} {\bf 39} (2014), 259-273.

\bibitem{motrad} D. Motreanu, V. R\u adulescu, Existence results for inequality problems with lack of convexity, {\it Numer. Funct. Anal. Optimiz.} {\bf 21} (2000), 869-884.

\bibitem{palais1} R. Palais, Ljusternik-Schnirelmann theory on Banach manifolds, {\it Topology} {\bf 5} (1960), 115-132.

\bibitem{palais2} R. Palais, S. Smale, A generalized Morse theory, {\it Bull. Amer. Math. Soc.} {\bf 70} (1964), 165-171.

\bibitem{pucci1} P. Pucci, J. Serrin, Extensions of the mountain pass theorem, {\it J. Funct. Anal.} {\bf 59} (1984), 185-210.

\bibitem{pucci2} P. Pucci, J. Serrin, A mountain pass theorem, {\it J. Differential Equations} {\bf 60} (1985), 142-149.
    
\bibitem{pucciz} P. Pucci, Q. Zhang, Existence of entire solutions for a
class of variable exponent elliptic equations, {\it J. Differential Equations}
{\bf 257} (2014), 1529-1566.

\bibitem{radulescu} V. R\u adulescu, Critical point theory, in {\it Selected Papers of James Serrin, vol. 2} (P.~Pucci, V.D.~R\u adulescu, H.~Weinberger, Eds.), Contemporary Mathematicians, Springer, Basel, 2014, pp.~431-433 (available at {\tt http://www.math.ucv.ro/$\sim$radulescu/articles/serrinvol13.pdf}).

\bibitem{radnla} V.D. R\u adulescu, Nonlinear elliptic equations with variable exponent: old and new, {\it Nonlinear Analysis: Theory, Methods and Applications} {\bf 121} (2015), 336-369.

\bibitem{radrepna} V.D. R\u adulescu, D. Repov\v{s}, Combined effects in nonlinear problems arising in the study of anisotropic continuous media, {\it Nonlinear Anal.} {\bf 75} (2012), 1524-1530.

\bibitem{radrep} V.D. R\u adulescu, D. Repov\v{s}, {\it  Partial Differential Equations with Variable Exponents: Variational Methods and Qualitative Analysis}, CRC Press, Taylor \& Francis Group, Boca Raton FL, 2015.

\bibitem{repovsaa} D. Repov\v{s}, Stationary waves of Schr\"{o}dinger-type equations with variable exponent, {\it Anal. Appl. (Singap.)} {\bf 13} (2015), 645-661.

\bibitem{yuc} Z. Y\"{u}ceda\u{g}, Solutions of nonlinear problems involving $p(x)$-Laplacian operator, {\it Adv. Nonlinear Anal.} {\bf 4} (2015),  285-293.

}
\end{thebibliography}



%%% ----------------------------------------------------------------------

\begin{abstract}
We study two classes of nonhomogeneous elliptic problems with Dirichlet boundary condition and involving a fourth order differential operator with variable exponent and power-type nonlinearities. The first result of this paper establishes the existence of a nontrivial weak solution in the case of a small perturbation of the right-hand side.
The proof combines variational methods, including the Ekeland variational principle and the mountain pass theorem of Ambrosetti and Rabinowitz.
Next we consider a very related eigenvalue problem and we prove the existence of nontrivial weak solutions for large values of the parameter.
 The direct method of the calculus of variations, estimates of the levels of the associated energy functional and basic properties of the Lebesgue and Sobolev spaces with variable exponent  have an important role in our arguments.
\end{abstract}
%%% ----------------------------------------------------------------------
\maketitle


\section{Introduction}
In a pioneering paper, A.~Ambrosetti, H.~Brezis and G.~Cerami \cite{abc} initiated the qualitative analysis of semilinear Dirichlet elliptic problems that involve concave and convex nonlinearities. They proved several existence, multiplicity and nonexistence results and developed powerful topological and variational methods for the study of such nonlinear problems. In particular, they studied the effects of small perturbations for the existence of solutions. In \cite{mrpams} and \cite{kong} there are established related existence results in the case of elliptic problems with variable exponents and Dirichlet boundary condition (see \cite{radnla} and \cite{radrep} for further developments and related properties). The main purpose of this paper is to complete the results in Kong \cite{kong}
and to prove the existence of a family of eigenvalues in a neighborhood of the origin.
We also refer to the related papers \cite{ferrara,molrep,radrepna,repovsaa,yuc}. Additional results on higher-order problems or nonlinear partial differential equations with variable exponent can be found in the papers by G.~Autuori, F.~Colasuonno, and P.~Pucci \cite{autuori}, Z.~Chen \cite{chen},
F.~Colasuonno and P.~Pucci \cite{cola}, A.~Kratohvil and I.~Necas \cite{krat}, V.~Lubyshev \cite{luby}, P.~Pucci and Q.~Zhang \cite{pucciz}.

Let $\Omega\subset\RR^N$ be a bounded domain with smooth boundary. Consider the following nonhomogeneous eigenvalue problem
\begin{equation}\label{1}
\left\{\begin{array}{lll}
&\di -\dpx u=\lambda |u|^{q(x)-2}u,&\quad x\in\Omega\\
&\di u=0,&\quad x\in\partial\Omega,
\end{array}\right.
\end{equation}
where $p,q:\overline\Omega\ri\RR$ are continuous functions and $\dpx$ denotes the {\it $p(x)$-Laplace operator}, which is defined by
$$\dpx u:={\rm div}\, (|\nabla u|^{p(x)-2}\nabla u).$$

Problem \eqref{1} is studied in \cite{mrpams} (see also \cite[Chapter 2.3.1]{radrep}) in a subcritical setting under the basic assumption
$$1<\min_{x\in\overline\Omega}q(x)<\min_{x\in\overline\Omega}p(x)<\max_{x\in\overline\Omega}q(x).$$
Under this hypothesis, the main result in \cite{mrpams} establishes that there exists $\lambda^*>0$ such that problem \eqref{1} has at least one nontrivial solution for all $\lambda\in(0,\lambda^*)$. Since the associated energy functional does not have a mountain pass geometry (see A.~Ambrosetti and P.~Rabinowitz \cite{ambrab}), the proof relies essentially on the Ekeland variational principle, see I.~Ekeland \cite{eke}. We point out that the original proof of the mountain pass theorem is based on several powerful deformation techniques developed by R.~Palais and S.~Smale \cite{palais1,palais2}, who developed the main ideas of the Morse theory into the abstract framework of the differential topology on infinite-dimensional Riemann manifolds. A simpler proof of the mountain pass theorem is due to H.~Brezis and L.~Nirenberg \cite{brenir}, who used a pseudo-gradient lemma, a perturbation argument and the Ekeland variational principle.

The study initiated in \cite{mrpams} and \cite{radrep} was continued in L.~Kong \cite{kong} in the framework of the {\it $p(x)$-biharmonic operator} $\dpx^2$, namely
$$\dpx^2u:=\Delta(|\Delta u|^{p(x)-2}\Delta u).$$

 Consider the fourth order nonlinear elliptic equation with variable exponent and Dirichlet boundary condition
\begin{equation}\label{1four}
\left\{\begin{array}{lll}
&\di \dpx^2 u+a(x)|u|^{p(x)-2}u=\lambda w(x)f(u),&\quad x\in\Omega\\
&\di u=\Delta u=0,&\quad x\in\partial\Omega,
\end{array}\right.
\end{equation}
where $a(x)$ and $w(x)$ are nonnegative potentials and the nonlinear term $f$ behaves like
$$f(u)=|u|^{\gamma(x)-2}u-|u|^{\beta(x)-2}u,$$
where $\gamma,\ \beta>1$ are continuous functions and under the basic hypothesis
\begin{equation}\label{ip}
\gamma (x)<\beta (x)<p(x),\quad\mbox{for all}\ x\in\overline\Omega.
\end{equation}
The main result in \cite{kong} asserts  that there exists $\lambda^*>0$ such that problem \eqref{1four} has at least one nontrivial solution for all $\lambda\in(0,\lambda^*)$.

In the present paper, we establish several existence results for  problems related to \eqref{1four} but under some basic assumptions different from \eqref{ip}.

We consider the following nonlinear problem
\begin{equation}\label{1pb}
\left\{\begin{array}{lll}
&\di \dpx^2 u+a|u|^{p(x)-2}u=\lambda (|u|^{\gamma(x)-2}-|u|^{\beta(x)-2})u,&\quad x\in\Omega\\
&\di u=\Delta u=0,&\quad x\in\partial\Omega,
\end{array}\right.
\end{equation}
where $\lambda$ is a positive parameter and $a\geq 0$. Under two different assumptions, we show that problem \eqref{1pb} has at least one nontrivial solution if the positive parameter $\lambda$ is {\it small enough}. The proof relies on the Ekeland variational principle and the mountain pass theorem.
We refer to J.~Garcia Azorero and I.~Peral Alonso \cite{garcia} who applied the mountain pass theorem to obtain the existence of a nodal (that is, sign-changing) solution in a related quasilinear setting.

The situation changes if we consider a problem very close to \eqref{1pb}. Let us consider the following eigenvalue nonlinear Dirichlet problem
\begin{equation}\label{1pbb}
\left\{\begin{array}{lll}
&\di \dpx^2 u+a|u|^{p(x)-2}u=\lambda |u|^{\gamma(x)-2}-|u|^{\beta(x)-2}u,&\quad x\in\Omega\\
&\di u=\Delta u=0,&\quad x\in\partial\Omega.
\end{array}\right.
\end{equation}
In this case, we establish a sufficient condition for the existence of nontrivial solutions provided that the parameter $\lambda$ is {\it large enough}. The proof is based on the direct method of the calculus of variations.

In the following section
we recall some basic definitions and properties concerning the basic function  spaces with variable exponent.
We refer to the recent monographs of L.~Diening, P.~H\"asto,  P.~Harjulehto and M.~Ruzicka \cite{die} and V.~R\u adulescu and D.~Repov\v{s} \cite{radrep} for related properties of Lebesgue and Sobolev spaces with variable exponents. 
The main results are stated in Section 3 of this paper. Final comments and some open problems are included in the last part of this paper.




\section{Function spaces with variable exponent}

Consider the set
$$C_+(\overline\Omega)=\{p\in C(\overline\Omega),\;p(x)>1\;{\rm
for}\;
{\rm all}\;x\in\overline\Omega\}.$$
For all $p\in C_+(\overline\Omega)$ we define
$$p^+=\sup_{x\in\Omega}p(x)\qquad\mbox{and}\qquad p^-=
\inf_{x\in\Omega}p(x).$$


For any $p\in C_+(\overline\Omega)$, we define the {\it variable exponent
Lebesgue space}
$$L^{p(x)}(\Omega)=\left\{u;\ u\ \mbox{is
 measurable and }
\int_\Omega|u(x)|^{p(x)}\;dx<\infty\right\}.$$
This vector space is a Banach space if it is endowed with the {\it Luxemburg norm}, which is defined by
$$|u|_{p(x)}=\inf\left\{\mu>0;\;\int_\Omega\left|
\frac{u(x)}{\mu}\right|^{p(x)}\;dx\leq 1\right\}.$$
The function space $L^{p(x)}(\Omega)$ is reflexive if and only if $1 < p^-\leq p^+<\infty$
 and continuous functions with compact support
are dense in $L^{p(x)}(\Omega)$ if $p^+<\infty$.

Let $L^{q(x)}(\Omega)$ denote the conjugate space
of $L^{p(x)}(\Omega)$, where $1/p(x)+1/q(x)=1$. If
$u\in L^{p(x)}(\Omega)$ and $v\in L^{q(x)}(\Omega)$ then  the following H\"older-type inequality holds:
\begin{equation}\label{Hol}
\left|\int_\Omega uv\;dx\right|\leq\left(\frac{1}{p^-}+
\frac{1}{q^-}\right)|u|_{p(x)}|v|_{q(x)}\,.
\end{equation}
Moreover, if $p_j\in C_+(\overline\Omega)$ ($j=1,2,3$) and
$$\frac{1}{p_1(x)}+\frac{1}{p_2(x)}+\frac{1}{p_3(x)}=1$$
then for all $u\in L^{p_1(x)}(\Omega)$, $v\in L^{p_2(x)}(\Omega)$, $w\in L^{p_3(x)}(\Omega)$
\begin{equation}\label{Hol1}
\left|\int_\Omega uvw\;dx\right|\leq\left(\frac{1}{p_1^-}+
\frac{1}{p_2^-}+\frac{1}{p_3^-}\right)|u|_{p_1(x)}|v|_{p_2(x)}|w|_{p_3(x)}\,.
\end{equation}

The inclusion between Lebesgue spaces also generalizes the classical framework, namely if
$0 <|\Omega|<\infty$ and $p_1$, $p_2$ are variable exponents so that $p_1\leq p_2$
 in $\Omega$ then there exists the continuous embedding
$L^{p_2(x)}(\Omega)\hookrightarrow L^{p_1(x)}(\Omega)$.


If $k$ is a positive integer number and $p\in C_+(\overline\Omega)$, we define the variable exponent  Sobolev space by
$$
W^{k,p(x)}(\Omega)=\{u\in L^{p(x)}(\Omega):\; D^\alpha u\in
L^{p(x)} (\Omega),\ \mbox{for all}\ |\alpha|\leq k \}.
$$
Here $\alpha =(\alpha_1,\ldots ,\alpha_N)$ is a multi-index, $|\alpha|=\sum_{i=1}^N\alpha_i$ and
$$D^\alpha u=\frac{\partial^{|\alpha|}u}{\partial^{\alpha_1}_{x_1}\ldots \partial^{\alpha_N}_{x_N}}\,.$$

On $W^{k,p(x)}(\Omega)$ we  consider the following norm
$$
\|u\|_{k,p(x)}=\sum_{|\alpha|\leq k}|D^\alpha u|_{p(x)}.
$$
Then $W^{k,p(x)}(\Omega)$ is a reflexive and separable Banach space. Let $W^{k,p(x)}_0(\Omega)$
denote the closure of $C^\infty_0(\Omega)$ in $W^{k,p(x)}(\Omega)$.

Consider the function space $E$ defined by
$$E=W^{1,p(x)}_0(\Omega)\cap W^{2,p(x)}(\Omega).$$
Then $E$ is a separable and reflexive Banch space if it is equipped with the norm
$$\|u\|_E=\|u\|_{1,p(x)}+\|u\|_{2,p(x)}\,.$$
Cf. Kong \cite[p. 251]{kong}, the norms $\|u\|_E$ and $|\Delta u|_{p(x)}$ are equivalent.

If $a$ is a positive number, define for all $u\in E$
$$\|u\|_a=\inf\left\{\lambda>0;\ \intom\left(\left|\frac{\Delta u}{\lambda} \right|^{p(x)}+
a\left|\frac{ u}{\lambda} \right|^{p(x)}
 \right)dx\leq 1\right\}\,.$$
 Then $\|u\|_a$ is well-defined and it is a norm which is equivalent with the norms $\|u\|_E$ and $|\Delta u|_{p(x)}$ in $E$.

Let  $\varrho_a:E\ri\RR$ be the {\it modular function} defined by
\bb\label{rho}
\varrho_{a}(u)=\int_\Omega\big(|\Delta
u|^{p(x)}+a|u|^{p(x)}\big)\,dx.
\bbb

If $(u_n)$, $u\in E$ then the following
properties are true:
\begin{equation}\label{M4}
\|u\|_a>1\;\Rightarrow\;\|u\|_a^{p^-}\leq \varrho_{a}(u)
\leq\|u\|_a^{p^+}\,,
\end{equation}
\begin{equation}\label{M5}
\|u\|_a<1\;\Rightarrow\;\|u\|_a^{p^+}\leq \varrho_{a}(u)
\leq\|u\|_a^{p^-}\,,
\end{equation}
\begin{equation}\label{M6}
\|u_n-u\|_E\rightarrow 0\;\;\;\Leftrightarrow\;\;\;\varrho_{a}
(u_n-u)\rightarrow 0\,.
\end{equation}

Let $p^*(x)$ denote the {\it critical Sobolev exponent}, namely
$$p^*(x)=\left\{
\begin{array}{lll}
&\di\frac{Np(x)}{N-2p(x)}&\quad
\mbox{if $2p(x)<N$}\\
&\di  +\infty&\quad\mbox{if $2p(x)\geq N$}.
\end{array}\right.
$$
We point out that if $p,q\in C_+(\overline\Omega)$
and $q(x)<p^\star(x)$ for all $x\in\overline\Omega$ then the
embedding
$E\hookrightarrow L^{q(x)}(\Omega)$
is compact, see Kong \cite[Proposition 1.3]{kong}.


The variable exponent Lebesgue and Sobolev
spaces are generalizations of the classical Lebesgue and Sobolev spaces, replacing the constant exponent $p$ with an exponent function $p(\cdot )$. These  spaces have been the subject of constant
interest since the beginning of the 20th century both as function spaces with intrinsic interest and for
their applications to problems arising in nonlinear partial differential equations and the calculus of variations. We refer to the monographs \cite{cruz,die,radrep} for related properties of these spaces and their history.

\section{The main results and related properties}
We say that $\lambda$ is an {\it eigenvalue} of problem \eqref{1pb} is there exists $u\in E\setminus\{0\}$ such that for all $v\in E$
$$ \intom |\Delta u|^{p(x)-2}\Delta u\Delta vdx+a\intom |u|^{p(x)-2}uvdx= \lambda \intom(|u|^{\gamma(x)-2}uv-|u|^{\beta(x)-2}uv)dx.$$
If $\lambda$ is an eigenvalue of problem \eqref{1pb}, the corresponding function $u\in E\setminus\{0\}$ is a {\it weak solution} of problem \eqref{1pb}.

We study problem \eqref{1pb} under one of the  following hypotheses:
\begin{equation}\label{hyp}
1<\gamma (x)<\min\{p(x),\beta (x)\}<\max\{p(x),\beta (x)\}<p^*(x)\quad\mbox{for all}\ x\in\overline\Omega \end{equation}
or
\begin{equation}\label{hypbis}
1<\min\{p(x),\beta (x)\}<\max\{p(x),\beta (x)\}<\gamma (x)<p^*(x)\quad\mbox{for all}\ x\in\overline\Omega .\end{equation}

The energy functional associated to problem \eqref{1pb} is defined as follows:
$$\ee (u)=\intom\frac{1}{p(x)}\left(|\Delta u|^{p(x)}+a|u|^{p(x)}\right)dx-\lambda\intom
\left[\frac{|u|^{\gamma(x)}}{\gamma (x)}-\frac{|u|^{\beta(x)}}{\beta (x)}\right]dx,$$
for all $u\in E$.

Hypothesis \eqref{hyp} implies that $\ee$ is well-defined, of class $C^1$, and for all $v\in E$
$$\langle\ee'(u),v\rangle =\intom \left(|\Delta u|^{p(x)-2}\Delta u\Delta v+a|u|^{p(x)-2}uv\right)dx-
\lambda\intom \left(|u|^{\gamma (x)-2}-|u|^{\beta (x)-2}\right)uvdx.$$

The first result of this paper is the following.

\begin{theorem}\label{t1}
 Assume that one of the hypotheses \eqref{hyp} or \eqref{hypbis} is satisfied. Then there exists a positive number $\lambda^*$ such that for all $\lambda\in (0,\lambda^*)$ problem \eqref{1pb} has at least one nontrivial weak solution.
\end{theorem}

We are then concerned with the study of problem \eqref{1pbb}. We say that $\lambda$ is an {\it eigenvalue} of problem \eqref{1pbb} is there exists $u\in E\setminus\{0\}$ such that for all $v\in E$
$$ \intom |\Delta u|^{p(x)-2}\Delta u\Delta vdx+a\intom |u|^{p(x)-2}uvdx= \lambda \intom|u|^{\gamma(x)-2}uvdx-\intom |u|^{\beta(x)-2}uvdx.$$

\begin{theorem}\label{t2}
 Assume that the hypothesis \eqref{hyp}  is satisfied. Then there exists a positive number $\lambda^{**}$ such that for all $\lambda\in (\lambda^{**},\infty)$ problem \eqref{1pbb} has at least one nontrivial weak solution.
\end{theorem}


We point out that hypothesis \eqref{hyp} implies that problem \eqref{1pb} does not have a mountain pass geometry. More precisely, $\ee$ satisfies one of the geometric hypotheses of the mountain pass theorem, namely the existence of a ``mountain" between two prescribed ``villages". However, the second geometric assumption of the mountain pass theorem is not fulfilled because this ``valley" is close to the first ``village" and not across the chain of mountains, as requested by the mountain pass theorem.
For this reason the existence of the solution follows with different arguments and only for small perturbations (in terms of $\lambda$). An interesting {\it open problem} is to provide a complete description for {\it all} values of the positive parameter $\lambda$.

We remark that Theorem \ref{t1} establishes a property related to Theorem 2.1 in \cite{kong}.
However, our result is based on the assumption \eqref{hyp}, which is more general than the corresponding hypothesis (2.1) in \cite{kong}.

The proofs of Theorems \ref{t1} and \ref{t2} use some ideas developed in \cite{mrpams,radrep,radrepna} in the framework of $p(x)$-Laplace operators and extended in \cite{kong} to biharmonic operators with variable exponent.

\subsection{Existence of a mountain and of a village.}
We are first concerned with the proof of Theorem \ref{t1} if the hypothesis \eqref{hyp} is fulfilled.

We have $\ee(0)=0$. We first establish the following auxiliary property.

\begin{lemma}\label{l1}
There exists a positive number $\lambda^*$ such that for all $\lambda\in (0,\lambda^*)$ there are positive numbers $r$ and $\eta$ such that $\ee (u)\geq r$ for all $u\in E$ with $\|u\|=\eta$.
\end{lemma}

{\it Proof.} We  observe that
$$\begin{array}{ll}
\ee (u)&\di\geq\frac{1}{p^+}\intom\left(|\Delta u|^{p(x)}+a|u|^{p(x)}\right)dx-\frac{\lambda}{\gamma^-}\intom
|u|^{\gamma(x)}dx+\frac{\lambda}{\beta^+}\intom
|u|^{\beta(x)}dx\\
&\di = \frac{1}{p^+}\varrho_a(u)-\frac{\lambda}{\gamma^-}\intom
|u|^{\gamma(x)}dx+\frac{\lambda}{\beta^+}\intom
|u|^{\beta(x)}dx\\
&\di\geq \frac{1}{p^+}\varrho_a(u)-\frac{\lambda}{\gamma^-}\intom
|u|^{\gamma(x)}dx\,.\end{array}$$

Fix $\eta\in (0,1)$ and assume that $\|u\|_E=\eta$. Using relation \eqref{M5} we obtain
$$\ee (u)\geq \frac{1}{p^+}\|u\|_E^{p^+}-\frac{\lambda}{\gamma^-}\intom
|u|^{\gamma(x)}dx\,.$$

Since the embedding $E\hookrightarrow L^{\gamma(x)}(\Omega)$ is continuous, there exists $C_1>0$ such that for all $u\in E$
$$\ee (u)\geq \frac{1}{p^+}\|u\|_E^{p^+}-\lambda C_1\|u\|_E^{\gamma^-}=\frac{\eta^{p^+}}{p^+}-\lambda C_1\eta^{\gamma^-}\,.$$

Now, taking $\lambda^*$ sufficiently small, we deduce that for all $\lambda\in (0,\lambda^*)$ there exists $r>0$ such $\ee(u)\geq r$ for all $u\in E$ with $\|u\|_E=\eta$.

The proof of Lemma \ref{l1} is now complete. \qquad $\square$

\smallskip
Next, we establish the existence of a valley near the origin.

\begin{lemma}\label{l2}
There exist $v\in E$ and $t_0>0$ such that $\ee (tv)<0$ for all $t\in (0,t_0)$.
\end{lemma}

{\it Proof.} Fix $v\in E\setminus\{0\}$ such that $v\geq 0$.  For all $t\in (0,1)$ we have
$$\begin{array}{ll}
\ee (tv)&\di =\intom\frac{t^{p(x)}}{p(x)}\left(|\Delta v|^{p(x)}+av^{p(x)}\right)dx-\lambda\intom
\frac{t^{\gamma(x)}}{\gamma (x)}v^{\gamma (x)}dx+\lambda\intom
\frac{t^{\beta(x)}}{\beta (x)}v^{\beta (x)}dx\\
&\di\leq \frac{t^{p^-}}{p^-}\varrho_a(v)-\lambda\frac{t^{\gamma^+}}{\gamma^+}\intom v^{\gamma(x)}dx+ \lambda\frac{t^{\beta^-}}{\beta^-}\intom v^{\beta(x)}dx\\
&\di = C_1t^{p^-}+C_2t^{\beta^-}-C_3t^{\gamma^+},\end{array}$$
where $C_1$, $C_2$, $C_3$ are positive numbers.

Using hypothesis \eqref{hyp}, we deduce that $\ee (tv)<0$, provided that $t>0$ is sufficiently small. The proof of Lemma \ref{l2} is complete. \qquad $\square$

\subsection{A compactness condition versus a variational principle.} We recall that a sequence $(u_n)\subset E$ is a {\it Palais-Smale sequence} if
\bb\label{pss}
\ee (u_n)=
O(1)\ \mbox{and}\ \|\ee '(u_n)\|_{E^*}=o(1)\ \mbox{as}\ n\ri\infty .
\bbb

Since the right-hand side of equation \eqref{1pb} does not satisfy the Ambrosetti-Rabinowitz condition, we cannot deduce that $\ee$ satisfies the Palais-Smale condition, that is, any Palais-Smale sequence is relatively compact. However, we prove in what follows that there is a suitable {\it bounded} Palais-Smale
sequence that contains a strongly convergent subsequence.

Returning to Lemma \ref{l1}, we have
\begin{equation}\label{11}\inf_{u\in \partial B}\ee (u)\geq r>0,\end{equation}
where $$B:=\{u\in E;\ \|u\|_a<\eta\}\,.$$

By Lemma \ref{l2}, there exists $v\in E$ such that
\begin{equation}\label{12}\ee (tv)<0\quad \mbox{for all $t>0$ small enough}.\bbb

Set $$m:=\inf_{u\in\overline B}\ee (u).$$
Then $m$ is finite and using relation \eqref{12}, we deduce that $m<0$. By \eqref{11} it follows that
$$\inf_{u\in \partial B}\ee (u)-\inf_{u\in \overline B}\ee (u)>0.$$
Fix $\ep>0$ such
$$\ep <\inf_{u\in \partial B}\ee (u)-\inf_{u\in \overline B}\ee (u).$$
The functional $\ee$ restricted to the complete metric space $\overline B$ satisfies the hypotheses of the Ekeland variational principle. A straightforward computation as in \cite[p. 46-47]{radrep} shows that there exists a {\it bounded sequence} $(u_n)\subset B$ such that
\bb\label{111} \ee (u_n)\ri m\quad\mbox{and}\quad \|\ee'(u_n)\|_{E^*}\ri 0\quad\mbox{as}\ n\ri\infty.\bbb
So, up to a subsequence, we can assume that
$$u_n\rightharpoonup u_0\quad\mbox{in}\ E$$
$$u_n\ri u_0\quad\mbox{in}\ L^{\gamma (x)}(\Omega)$$
and
$$u_n\ri u_0\quad\mbox{in}\ L^{\beta (x)}(\Omega).$$
We claim that, in fact,
$$u_n\ri u_0\quad\mbox{in}\ E.$$

Using the second information in relation \eqref{111} we deduce that for all $\varphi\in E$
\bb\label{113}\begin{array}{ll}
&\di\intom \left[|\Delta u_n|^{p(x)-2}\Delta u_n\Delta (u_n-u_0)+a|u_n|^{p(x)-2}u_n(u_n-u_0)\right]dx-\\ &\di
\lambda\intom \left(|u_n|^{\gamma (x)-2}-|u_n|^{\beta (x)-2}\right)u_n(u_n-u_0)dx\ri 0\end{array}\bbb
as $n\ri\infty$.

By \cite[Lemma 2.1(b)]{kong}, the operator $\ee' :E\ri E^*$ is an operator of type $(S_+)$. Thus
we obtain that $u_n\ri u_0$ in $E$, which is our claim. So, by \eqref{111},
$$\ee (u_0)=m<0\quad\mbox{and}\quad \ee'(u_0)=0.$$
We conclude that $u_0$ is a nontrivial weak solution of problem \eqref{1pb}. Thus all $\lambda\in (0,\lambda^*)$ is an eigenvalue of problem \eqref{1pb}. The proof of Theorem \ref{t1} is now complete, provided that hypothesis \eqref{hyp} is fulfilled.

\smallskip
We are now concerned with the related property if condition \eqref{hypbis} is satisfied. We first observe that under this new hypothesis, the conclusion of Lemma \ref{l1} remains true. Next, since condition \eqref{hypbis} implies that the dominating term in the right-hand side of problem \eqref{1pb} is $|u|^{\gamma(x)-2}u$, we prove in what follows the existence of a valley {\it across} the chain of mountains.

\begin{lemma}\label{l3}
There exist $v\in E$ and $t_0>0$ such that $\ee (tv)<0$ for all $t>t_0$.
\end{lemma}

{\it Proof.} Fix $v\in E\setminus\{0\}$ such that $v\geq 0$. By \eqref{M4} we deduce that for all $t>1$ we have
$$\begin{array}{ll}
\ee (tv)&\di =\intom\frac{t^{p(x)}}{p(x)}\left(|\Delta v|^{p(x)}+av^{p(x)}\right)dx-\lambda\intom
\frac{t^{\gamma(x)}}{\gamma (x)}v^{\gamma (x)}dx+\lambda\intom
\frac{t^{\beta(x)}}{\beta (x)}v^{\beta (x)}dx\\
&\di\leq \frac{t^{p^+}}{p^-}\varrho_a(v)-\lambda\frac{t^{\gamma^-}}{\gamma^+}\intom v^{\gamma(x)}dx+ \lambda\frac{t^{\beta^+}}{\beta^-}\intom v^{\beta(x)}dx\\
&\di = C_4t^{p^-}+C_5 t^{\beta^+}-C_6t^{\gamma^-},\end{array}$$
where $C_4$, $C_5$, $C_6$ are positive numbers.

Using hypothesis \eqref{hyp}, we
have $\gamma^->\max\{p^-,\beta^+\}$. It follows that
  $\ee (tv)<0$, provided that $t>0$ is sufficiently large. The proof of Lemma \ref{l3} is complete. \qquad $\square$

\subsection{Verification of the Palais-Smale condition}
We recall that the energy functional $\ee:E\ri\RR$ satisfies the Palais-Smale condition if any sequence $(u_n)\subset E$ such that
\bb\label{ps}
\ee (u_n)=
O(1)\ \mbox{and}\ \|\ee '(u_n)\|_{E^*}=o(1)\ \mbox{as}\ n\ri\infty ,
\bbb
is relatively compact.

Let $(u_n)\subset E$ be a sequence such that relation \eqref{ps} is fulfilled. We claim that
\bb\label{claim}
(u_n)\ \mbox{is bounded in}\ E .
\bbb

Arguing by contradiction, we suppose that the sequence $(u_n)$ is unbounded in $E$. Without loss of generality, we can assume that $\|u_n\|_a>1$ for all $n\geq 1$. Using relation \eqref{ps} we have
$$\begin{array}{ll}
O(1)+o(\|u_n\|)&\di =\ee (u_n)-\frac{1}{\gamma^-}\langle\ee'(u_n),u_n\rangle\\
&\di =\intom\frac{1}{p(x)}\left(|\Delta u_n|^{p(x)}+a|u_n|^{p(x)}\right)dx-\lambda\intom
\left[\frac{|u_n|^{\gamma(x)}}{\gamma (x)}-\frac{|u_n|^{\beta(x)}}{\beta (x)}\right]dx\\
&\di -\frac{1}{\gamma^-}\intom \left(|\Delta u_n|^{p(x)}+a|u_n|^{p(x)}\right)dx-
\frac{\lambda}{\gamma^-}\intom \left(|u_n|^{\gamma (x)}-|u_n|^{\beta (x)}\right)dx.
\end{array}$$
By relation \eqref{M4} we deduce that
$$\begin{array}{ll}
O(1)+o(\|u_n\|)&\di =\ee (u_n)-\frac{1}{\gamma^-}\langle\ee'(u_n),u_n\rangle\\
&\di\geq\left(\frac{1}{p^+}-\frac{1}{\gamma^-}\right)\intom\left(|\Delta u_n|^{p^-}+a|u_n|^{p^-}\right)dx\\
&\di +\lambda \intom \left(\frac{1}{\beta(x)}-\frac{1}{\gamma^-}\right)|u_n|^{\beta (x)}dx
+\lambda \intom \left(\frac{1}{\gamma^-}-\frac{1}{\gamma (x)}\right)|u_n|^{\gamma (x)}dx.
\end{array}$$
Using now the hypothesis \eqref{hypbis} we conclude that
$$O(1)+o(\|u_n\|)\geq \left(\frac{1}{p^+}-\frac{1}{\gamma^-}\right)\|u_n\|_a^{p^-}\quad\mbox{as}\ n\ri\infty.$$
Since $\gamma^->p^+$ it follows that
$$\|u_n\|_a=O(1)\quad\mbox{as}\ n\ri\infty.$$
This shows that our claim \eqref{claim} is true. So, up to a subsequence, we can assume that
$$u_n\rightharpoonup u_0\quad\mbox{in}\ E$$
$$u_n\ri u_0\quad\mbox{in}\ L^{\gamma (x)}(\Omega)$$
and
$$u_n\ri u_0\quad\mbox{in}\ L^{\beta (x)}(\Omega).$$
We show in what follows that
$$u_n\ri u_0\quad\mbox{in}\ E.$$

Using the second information in relation \eqref{ps} we deduce that for all $\varphi\in E$
$$\begin{array}{ll}
&\di\intom \left[|\Delta u_n|^{p(x)-2}\Delta u_n\Delta \varphi+a|u_n|^{p(x)-2}u_n\varphi\right]dx-\\ &\di
\lambda\intom \left(|u_n|^{\gamma (x)-2}-|u_n|^{\beta (x)-2}\right)u_n\varphi dx\ri 0\end{array}$$
as $n\ri\infty$.

With the same arguments as in the first case and since  $\ee' :E\ri E^*$ is an operator of type $(S_+)$, we conclude that
 $u_n\ri u_0$ in $E$, which shows that Palais-Smale condition is satisfied. At this stage it is enough to apply the mountain pass theorem in order to obtain a nontrivial weak solution of problem \eqref{1pb} for all $\lambda>0$, provided that the condition \eqref{hypbis} is satisfied.

 The proof of Theorem \ref{t1} is complete.
\qquad $\square$

\subsection{Proof of Theorem \ref{t2}.} The energy functional associated to problem \eqref{1pbb} is defined as follows:
$$\jj (u)=\intom\frac{1}{p(x)}\left(|\Delta u|^{p(x)}+a|u|^{p(x)}\right)dx-\lambda\intom
\frac{|u|^{\gamma(x)}}{\gamma (x)}dx+\intom\frac{|u|^{\beta(x)}}{\beta (x)}dx,$$
for all $u\in E$.

We show that $\jj$ is coercive, namely
$$\jj (u)\ri +\infty\quad\mbox{as}\ \|u\|_a\ri\infty.$$
Indeed, for all $u\in E$ with $\|u\|_a>1$ we have
$$\begin{array}{ll}
\jj (u)&\di\geq\frac{1}{p^+}\intom \left(|\Delta u|^{p^-}+a|u|^{p^-}\right)dx-\frac{\lambda}{\gamma^-}\intom
|u|^{\gamma(x)}dx+\frac{1}{\beta^+}\intom
|u|^{\beta(x)}dx\\
&\di\geq\frac{1}{p^+}\|u\|_a^{p^-}-\frac{\lambda}{\gamma^-}\intom
|u|^{\gamma(x)}dx\\
&\di\geq \frac{1}{p^+}\|u\|_a^{p^-}-\frac{c\lambda}{\gamma^-}\|u\|_a^{\gamma^+},
\end{array}$$
where $c$ is the best constant of the continuous embedding $E\hookrightarrow L^{\gamma(x)}(\Omega)$.
By hypothesis \eqref{hyp} we have $p^->\gamma^+$, which infers that the energy functional $\jj$ is coercive.

Let $(v_n)$ be a minimizing sequence of the functional $\jj$ in $E$. Since $\jj$ is coercive, we deduce that $(v_n)$ is a bounded sequence. So, up to a subsequence, we can assume that
$$v_n\rightharpoonup v_0\quad\mbox{in}\ E$$
$$v_n\ri v_0\quad\mbox{in}\ L^{\gamma (x)}(\Omega)$$
and
$$v_n\ri v_0\quad\mbox{in}\ L^{\beta (x)}(\Omega).$$
Using now the lower semicontinuity of $\jj$ (see \cite[Lemma 2.1 (a)]{kong}) we deduce that $v_0$ is a global minimizer of $\jj$ on $E$. It remains to prove that $v_0\not=0$. We have $\jj(0)=0$. Thus it is enough to show that
$$\inf\{\jj(v);\ v\in E\}<0\quad\mbox{for $\lambda$ big enough}.$$

Indeed, let us consider the following constrained minimization problem
\bb\label{constr}\lambda^{**}:=\inf\left\{\intom\frac{1}{p(x)}\left(|\Delta w|^{p(x)}+a|w|^{p(x)}\right)dx+
\intom\frac{|w|^{\beta(x)}}{\beta (x)}dx;\ w\in E\ \mbox{and}\ \intom
\frac{|w|^{\gamma(x)}}{\gamma (x)}dx=1\right\}.\bbb
If $(w_n)\subset E$ is an arbitrary minimizing sequence of problem \eqref{constr} then $(w_n)$ is bounded. Thus, up to subsequence, $(w_n)$ converges weakly in $E$ and strongly in $L^{\gamma(x)}(\Omega)$ and $L^{\beta (x)}(\Omega)$ to some $w_0$ satisfying
$$\intom
\frac{|w_0|^{\gamma(x)}}{\gamma (x)}dx=1$$and
$$\lambda^{**}=\intom\frac{1}{p(x)}\left(|\Delta w_0|^{p(x)}+a|w_0|^{p(x)}\right)dx
+\intom\frac{|w_0|^{\beta(x)}}{\beta (x)}dx>0.$$
We conclude that
$$\jj(w_0)=\lambda^{**}-\lambda<0\quad\mbox{for all $\lambda>\lambda^{**}$},$$
hence $w_0$ is a nontrivial weak solution of problem \eqref{1pbb}.
The proof of Theorem \ref{t2} is complete.
\begin{flushright}$\square$\end{flushright}


\subsection{Final comments.}
The analysis of the proofs of Theorems \ref{t1} and \ref{t2} shows that the results remain true if the left-hand side of problems \eqref{1pb} and \eqref{1pbb} is replaced with
$$\dpx^2 u+\alpha |u|^{p(x)-2}u,$$
where $\alpha$ is a real number such that the operator $\dpx^2 u+\alpha |u|^{p(x)-2}u$ is {\it coercive} in $E$, hence there is some $C>0$ such that for all $u\in E$
$$\intom\left(|\Delta u|^{p(x)}+\alpha |u|^{p(x)}\right)dx\geq C\varrho_{a}(u).$$
Even more, we expect that the results established in this paper are true for more general operators, say Leray-Lions operators with variable exponents. We refer here to the pioneering paper of J.~Leray and J.-L.~Lions \cite{leray}.

The existence properties established in Theorems \ref{t1} and \ref{t2} remain valid if the {\it bounded} domain $\Omega$ is replaced with an {\it unbounded} domain with boundary $\partial\Omega$. In such a case there are used local arguments, see F.~Gazzola and V.~R\u adulescu \cite[p. 59]{gazrad}.

We point out that the  results of this paper can be extended in a nonsmooth multi-valued setting, namely under weaker assumptions on the right-hand side of problems \eqref{1pb} and \eqref{1pbb}, which imply that the associated energy functionals are no longer of class $C^1$. This corresponds to {\it variational-hemivariational inequalities}. We refer to  D.~Motreanu and V.~R\u adulescu \cite{motrad} for a related inequality problem.

 Problems \eqref{1pb} and \eqref{1pbb} have been studied in this paper in the subcritical case, which corresponds to the basic assumption that the growth of the variable exponents $\beta$, $\gamma$ and $p$ is inferior than the critical exponent $p^*(x)$ for all $x\in\overline\Omega$. This hypothesis is crucial in order to ensure related compact embeddings of $E$ into Lebesgue spaces with variable exponent. A very interesting {\it open problem} is to study problems \eqref{1pb} and \eqref{1pbb} in the following {\it almost critical} setting: there exists $x_0\in\Omega$ such that
$$\max\{p(x),\beta (x),\gamma (x)\}<p^*(x)\ \mbox{for all $x\in\overline\Omega\setminus\{x_0\}$}\ \mbox{and}\ \max\{p(x_0),\beta (x_0), \gamma (x_0)\}=p^*(x_0).$$

We believe that a very interesting research subject is to study  problems \eqref{1pb} and \eqref{1pbb} if the biharmonic operator with variable exponent $\dpx^2u$
is replaced with an operator with  several variable exponents, for instance
$$\Delta\left((|\Delta u|^{p_1(x)-2}+|\Delta u|^{p_2(x)-2})\Delta u\right).$$

We conclude with a very interesting {\it open problem} concerning \eqref{1pb} under the hypothesis \eqref{hypbis}. We have applied in our proof the standard mountain pass theorem of A.~Ambrosetti and P.~Rabinowitz \cite{ambrab}. This pioneering result corresponds to mountains of {\it positive altitude}. The degenerate case is associated with mountains of {\it zero altitude} and was established by P.~Pucci and J.~Serrin \cite{pucci1,pucci2} (see also R\u adulescu \cite{radulescu} for an overview of these results). We suggest to formulate the optimal assumptions for the right-hand side of equation \eqref{1pb} in order to study this problem in the degenerate case of mountains of zero altitude.



\medskip
{\bf Acknowledgments.} This project was funded by the National Plan of Sciences, Technology and Innovation (MAARIFAH), King Abdulaziz City for Sciences and Technology, Kingdom of Saudi
Arabia, Award number (12-MAT2912-02).



\end{document}

